**Author's response to reviews**

**Title:** Increased serum levels of tumour-associated trypsin inhibitor independently predict a poor prognosis in colorectal cancer patients

**Authors:**

- Alexander Gaber (alexander.gaber@med.lu.se)
- Björn Nodin (bjorn.nodin@skane.se)
- Kristina Hotakainen (kristina.hotakainen@hus.fi)
- Elise Nilsson (elise.nilsson@med.lu.se)
- Ulf-Håkan Stenman (ulf-hakan.stenman@helsinki.fi)
- Anders Bjartell (anders.bjartell@med.lu.se)
- Helgi Birgisson (helgi.birgisson@surgsci.uu.se)
- Karin Jirstörm (karin.jirstrom@med.lu.se)

**Version:** 6 **Date:** 17 September 2010

**Author's response to reviews:**

Dear Editor,

Everything looks fine now regarding the text and figures. However, we noticed that reference no 27 had accidentally been added as nr 8. This has now been corrected and all references doublechecked again.

Warm regards,

Karin Jirström